Author’s response to reviews

Title: Study protocol for a randomized, blinded, controlled trial of ketamine for acute painful crisis of sickle cell disease

Authors:

Mohammed Alshahrani (msshahrani@iau.edu.sa)
Laila Perlas-Asonto (lasonto@iau.edu.sa)
Mohammed Eltahan (Mohamedrefaateltahan@yahoo.com)
Amal Alsulaibikh (asulaibeekh@iau.edu.sa)
Sukayna Alfaraj (Sukayna_alfaraj@hotmail.com)
Abdullah Almulhim (ABAMulhim@iau.edu.sa)
Murad Alabbad (Malabbad@iau.edu.sa)
Samar Alnahhash (dr.samaralnahash@yahoo.com)
Moath Aldarweesh (darwish.m85@gmail.com)
Alaa Mahmoud (badr_surg_2008@yahoo.com)
Nisreen Almaghraby (NHMaghraby@iau.edu.sa)
Mohammed Aljumaan (majumaan@iau.edu.sa)
Thamir Aljunaid (dr.tjunaid@hotmail.com)
Faisal Alhawaj (hawaj10@gmail.com)
Samar Alkenany (dr.smsmer@hotmail.com)
Omaima Elsayed (omaimafathy@yahoo.com)
Haitham Abdelwahab (doctor_sameer2003@yahoo.com)
Mohamed Moussa (dr.mousa80@yahoo.com)
Duaa Alsulaiman (Duaa.alsulaiman@gmail.com)
Bader Alosaimi (bader_k_o@hotmail.com)
Author’s response to reviews:

TRLS-D-18-00656R1

Study protocol for a randomized, blinded, controlled trial of ketamine for acute painful crisis of sickle cell disease

Oscar Bortolami, Ph.D,
Trials
https://trialsjournal.biomedcentral.com/Editor-in-Chief

Dear Dr. Bortolami,

Thank you for considering our manuscript “Study protocol for a randomized, blinded, controlled trial of ketamine for acute painful crisis of sickle cell disease” for review and potential publication in Trials Journal.

Please find attached specific reply to the minor concerns raised along with the revised manuscript.

Thank you for the opportunity for considering our revised manuscript for publication in The Trials Journal.
Sincerely,
On behalf of all authors

Mohammed Alshahrani, M.D.
Consultant, Emergency and Critical Care Medicine
Director, Critical Care Medicine Department
Vice Dean for Hospital Affairs & Director General
King Fahad Hospital of the University
Imam Abdulrahman Bin Faisal University-Dammam
PO Box 40236, Al Khobar Kingdom of Saudi Arabia
+966-556-966663; +966-3-896-6666 ext. 2010 (office)
Email: msshahrani@iau.edu.sa

Editor’s Comments:

1. Please do not disclose block size in the protocol.
   Response:
   Thank you for this comment. Correction was made in the revised manuscript.

2. Can you please state which software was used to calculate the boundaries?
   Reply:
   Thank you for this observation. The comments on interim analysis was answered and inserted in the revised protocol. We used IBM SPSS Statistics V25.